Author's response to reviews

Title: Transdermal rotigotine for the perioperative management of restless legs syndrome

Authors:

Birgit Högl (birgit.ho@i-med.ac.at)
Wolfgang H Oertel (oertelw@med.uni-marburg.de)
Erwin Schollmayer (erwin.schollmayer@ucb.com)
Lars Bauer (lars.bauer@ucb.com)

Version: 3 Date: 24 July 2012

Author's response to reviews:

We addressed the comments of reviewer 2 and changed the manuscript:
1. Abstract pg 2, background: ‘suitable’ was replaced by ‘feasible’
2. Abstract pg 2, conclusions: the last sentence was removed
3. Main text pg 11, the words ‘suitable and’ were removed

We hope that the revised manuscript now meets the requirements for publication.